Reply to REVIEWER 1 (05296794):
1. Thank for your review and important suggestion.
2. In the core tip, the sentence of “our patient is the first case report of solitary fibrous tumor of the rectal mesentery with postoperative recurrence and sarcomatosis.” has been reorganized.
3. In the part of history of present illness, the immunohistochemical result of the origin SFT pathology has been added.
4. I have added a reference to the sentence “Local recurrence or metastasis develops in 12%–22% of cases.” in page 6.
5. About use of FDG-PET to verify the tumor or detect metastasis, information has been revised in page 8.
6. Information about use of type-B ultrasonic and CT to find SFTs has been added in pages 8 and 9.
7. References of 27, 28, 29, 30 and 44 are added to support the revision above.
8. The draft has been revised by a native English again.

Reply to REVIEWER 2 (05260869):
1. Thank for your review and important suggestion.
2. The temporal terms "4 years ago", "suspected recently" etc. are revised into actual dates.
3. About the previous surgery, the "white nodule found in the peritoneal cavity" has been described more clearly in page 4.
4. During the last 28 months, the follow-up process has been added in page 7.
5. The guidance of management of these rare/difficult cases after a recurrence has been added in page 12.
6. I have corrected the mis-typed “1-year follow-up period” into “28-month follow-up period”
7. About the pubmed terms of search to perform the review of similar cases, I have added the description in pages 7 and 8.
8. The description of “first resection of a SFT with malignant potential” has been added in page 4. Besides, description of our surgical pathology feature for suspected malignancy could have described more clearly in page 10.
9. The draft has been revised by a native English again.